Math 503: 4th problem set

John Baldwin

February 24, 1998

1. Show that if T is strongly minimal and a ∈ acl(b, c), a¬ ∈ acl(b) then c ∈ acl(a, b). (Hint: to do this by hammer and tongs you must use the uniform bound on the size of an algebraic set and move to an elementary extension of the original model.)

2. Prove that any strongly minimal set D has a basis X such that D is algebraic closure of X, X is algebraically independent and any two such bases have the same cardinality.

3. Conclude that the theory of a strongly minimal set is categorical in every uncountable cardinal. What goes wrong in ℵ_0?